

# NOTA DI LAVORO 111.2009

Health-enhancing Activities and the Environment: How Competition for Resources Makes the Environmental Policy Beneficial

By **Xavier Pautrel**, Institut d'Economie et de Management de Nantes-IAE, Université de Nantes

# SUSTAINABLE DEVELOPMENT Series Editor: Carlo Carraro

# Health-enhancing Activities and the Environment: How Competition for Resources makes the Environmental Policy Beneficial

By Xavier Pautrel, Institut d'Economie et de Management de Nantes-IAE, Université de Nantes

# Summary

In a two-period overlapping generations model, this paper demonstrates that the relationship between the environmental taxation and the economic activity (level- and growth-output) becomes inverted-U shaped, when the detrimental impact of pollution on health and the private decision of each working-age agent to improve her health are taken into account. Especially, a tighter environmental tax is more likely to promote (rather than to harm) output-level and –growth when health is very sensitive to pollution, the weight of health in preferences is high, the polluting capacity of the production technology is high and the rate of natural purification of pollutants is low. The inverted-U shaped relationship between the environmental tax and the economic activity is due to a positive effect arising from the competition for resources between the final output sector and the health-enhancing activities that offsets the conventional detrimental "drag-down effect" for low values of the environmental tax. We also demonstrate that the link between the environmental tax and the determinants of the optimal environmental tax.

Keywords: Growth, Environment, Health, Overlapping Generations

JEL Classification: Q5

Address for correspondence:

Xavier Pautrel Nantes Atlantique Université Laboratoire d'Économie de Nantes (len) Chemin de la Censive du Tertre BP 52231 44322 Nantes Cedex 3 France E-mail: xavier.pautrel@univ-nantes.fr

# Health-enhancing activities and the environment: How competition for resources make the environmental policy beneficial

Xavier Pautrel\* xavier.pautrel@univ-nantes.fr

April 9, 2009

#### Abstract

In a two-period overlapping generations model, this paper demonstrates that the relationship between the environmental taxation and the economic activity (level- and growth-output) becomes inverted-U shaped, when the detrimental impact of pollution on health and the private decision of each working-age agent to improve her health are taken into account. Especially, a tighter environmental tax is more likely to promote (rather than to harm) output-level and -growth when health is very sensitive to pollution, the weight of health in preferences is high, the polluting capacity of the production technology is high and the rate of natural purification of pollutants is low.

The inverted-U shaped relationship between the environmental tax and the economic activity is due to a positive effect arising from the competition for resources between the final output sector and the health-enhancing activities that offsets the conventional detrimental "drag-down effect" for low values of the environmental tax.

We also demonstrate that the link between the environmental tax and the lifetime welfare is inverted-U shaped as well. Finally, we investigate the social optimum and the determinants of the optimal environmental tax.

Keywords : Growth; Environment; Health; Overlapping generations.

<sup>&</sup>lt;sup>\*</sup>address: Nantes Atlantique Université, Laboratoire d'Économie de Nantes (LEN), Chemin de la Censive du Tertre, BP 52231, 44322 Nantes Cedex 3, France.

# 1 Introduction

Is environmental policy harmful to economic activity, both in terms of output level and growth? Does the reduction of pollution imply a cost for economic activities so heavy that the gains from a better environment quality are not able to offset it? At the theoretical level, the answers are not clear-cut.

The aim of this article is to contribute to the debate focusing on one of the more striking features of pollution: its detrimental impact on health. Conversely to previous works in the field that take into account the impact of pollution on life expectancy (i.e. mortality), we focus our attention on the influence of pollution on illness and disability (morbidity). Indeed, a growing set of empirical evidence find a link between pollution and chronic diseases such as cancer, diabetes, hypertension, stroke, heart disease, pulmonary conditions and mental disorders, amongst others. Even if pollution is not the main source of these diseases, it contributes as a factor favouring their emergence or their worsening. Conversely to mortality that affects mainly the oldest old, illness and disability due to chronic diseases primarily impacts the working-age population, leading to important losses of productivity and rising health-expenditures mainly for the 30-50 old age people. Devol and Bedroussian (2007) from the Milken Institute estimate that the seven common chronic diseases represent for the United-States \$277 billion spent annually on treatment and a lost productivity equals to \$1.1 trillion per year.

The rising health-expenditures for working-age people and the time they have to devote in order to accommodate to their chronic disease creates a competition for resources that could be used in alternative ways, especially growth-led activities or final production activities.<sup>1</sup> The main contribution of this paper is to demonstrate that such a resource competition is a channel of transmission between the environmental policy, economic activity and welfare, when the detrimental impact of pollution on health status of working-age population is taken into account.

To demonstrate this point and analyze its implications, we use a two-period overlapping generations model in the veins of previous articles which addressed intergenerational environmental issues,<sup>2</sup> introducing an explicit link between the environment and health. Following empirical evidence, we assume that health is negatively influenced by pollution but is improved by the investment in health-enhancing activities made by each agent at her working-age.<sup>3</sup> Pollution is a by-product of final output production and in the competitive

<sup>&</sup>lt;sup>1</sup>Competition for resources in the relation between health and growth has been already studied by several articles (see Dormont et al., 2007, for details and references). Nevertheless, most of these contributions view better health as an increase in life expectancy. Empirically, Dormont et al. (2006) find changes in morbidity that induce savings which more than offset the increase in spending due to population ageing.

<sup>&</sup>lt;sup>2</sup>For example, John and Pecchenino (1994) who analyze the potential conflict between economic growth and the maintenance of environmental quality when consumption degrades environmental quality while investment in environmental maintenance promotes it. See also John et al. (1995) who investigate the effects of environmental taxation distinguishing the horizon of the agents and the economy. For models with non-renewable resources, see Agnani et al. (2005), Kemp and Long (1979), Mourmouras (1991, 1993).

<sup>&</sup>lt;sup>3</sup>Because we consider that agents are suffering from chronic diseases that required medical care when they are young we do not assume that the poor health agents expend more on medical care when they are elderly like Gutiérrez (2008). And, conversely to Williams (2002, 2003), we do not assume that ill agents do not work.

economy the government taxes final output to limit pollution emissions.

Our first contribution is to demonstrate that if the detrimental effect of pollution on health-status and an endogenous investment in health by the working-age are taken into account, the link between the environmental taxation and the economic activity (final output level and growth) is inverted-U shaped. Indeed, a tighter environmental policy has two opposite effects. First, because the environmental tax is imposed on final output, it reduces the rewards to production factor: the conventional "drag-down effect". Second, it reduces pollution and therefore improves health-status of the working-age. Agents reduce their investment in health-enhancing activities and the freed resources are used to increase consumption and production. This second effect, called "resource competition effect" is positive. Therefore, for low values of the environmental tax, the second effect offsets the first one, and the environmental policy promotes the level and the growth rate of output, as well as the global welfare.

Our second contribution is to show that the greater the room for an improvement in the health status, the more likely the environmental policy promotes economic activity and growth. That occurs when the rate of natural health decay is low, the efficiency of the health care spending is low, the weight of health in preferences is high, the part of labor in final output is high, the rate of natural purification of pollutants is low, the polluting capacity of production technology is high, the detrimental impact of pollution on health and the elasticity of pollution stock with respect to the net flow of pollution are high. Most of these criterions exist in the most developed countries, and because the detrimental impact of pollution on health is well-documented worldwide, our results show that an active environmental policy in these countries is highly probable to promote growth and output level: the positive gains in terms of health and growth should be higher than the losses from factor rewards.

Finally, we investigate the social optimum and the optimal environmental tax. We demonstrate that the higher the weight of health in preferences, the elasticity of pollution stock with respect to the net flow of pollution, the detrimental impact of pollution on health and/or the part of labor in production, the higher the optimal environmental tax is.

The paper is set out as follows. Section 2 gives empirical evidence on the link between pollution illness and disability. Section 3 presents the model. Section 4 studies the competitive equilibrium and the impact of the environmental taxation on the steady-state. Section 5 investigates the social optimum and the optimal environmental tax. Section 6 examines two extensions: AK endogenous growth and the introduction of the impact of health on labor productivity. Section 7 concludes.

# 2 Pollution, illness and disability

The major part of the environmental economic contributions integrating the detrimental impact of pollution on health focus on life expectancy.<sup>4</sup> Nevertheless, pollution also affects

<sup>&</sup>lt;sup>4</sup>Among others, see Jouvet et al. (2007), Mariani et al. (2008), Varvarigos (2008) or Pautrel (2008, 2009). The aforementioned articles by Gutiérrez (2008) and Williams (2002, 2003) are exceptions.

morbidity by favouring or worsening some chronic diseases that do not always lead to death but have durable detrimental impacts in terms of illness and disability.

These chronic diseases, that encompasse cancer, diabetes, hypertension, stroke, heart disease, asthma, obesity are a growing burden because they represent 60% of all deaths worldwide and they are a major source of disability (see WHO, 2004, 2005). While they strongly strike down developing countries, they also represent a great burden for developed economies, like the United-States, England, Canada, Israel, and Australia, as reported by Suhrcke et al. (2006b) and Zhang et al. (2008) amongst others. For example, Devol and Bedroussian (2007) find that more than half of all Americans (55.8%) suffered from one or more chronic diseases in 2003.

The causes of increasing the risk to develop chronic diseases are well-established and well-known: mainly unhealthy diet, physical inactivity and tobacco use. Nevertheless, recent studies emphasize the importance of the environmental factors in the development and the worsening of some of these chronic diseases, especially in the developed countries. According to Briggs (2003) about 8-9% of the total disease burden may be attributed to pollution in developed countries.<sup>5</sup>

In the case of air pollution for example, it is well-established that particulate matter pollution, Nitrogen dioxide, Sulfur dioxide and ozone favour the onset of asthma crisis and aggravate respiratory diseases, that carbon monoxide affects mental function... (for a study on European countries see Katsouyanni, 2003). Furthermore, particulate matter air pollution plays a role as a cause of the development (pathogenesis in medical terms) of cardiovascular disease (Brook et al., 2004) or lung cancer (Pope et al., 2002), and it may be particularly harmful to high-risks people with diabetes and people with hypertension (Pope and al., 2004). Moreover, air pollution could be deleterious to vascular health especially for people with diabetes (Rajagopalan et al., 2005; O'Neill et al., 2005). Water pollution [Paulu et al. (1999), Valent et al. (2004)] and industrial pollution [Nadal et al. (2004), Chen and Liao (2005), Schuhmacher and Domingo (2006)] are also reported as detrimental for health.

Lang et al. (2008) recently found that higher unary BPA concentration (chemical pollution) are associated with cardiovascular diagnoses and with diabetes for adults. Smink et al. (2008) show that children in the higher exposure group of HCB (a pesticide extensively used before it was banned from the United-States) had an increased risk of being overweight and obese, even if the mother is normalweight. The epidemiological association between persistent organic pollutants and diabetes has been also demonstrated by Rylander et al. (2005) and Porta (2006). Even if such an association does not prove necessarily a causal relation, as noted by Jones et al. (2008), such a causal link could be biologically explained (see Remillard and Bunce, 2002).

Furthermore, the link between obesity and diabetes seems to be related to pollution. Lee et al. (2007) find that the expected association between obesity and diabetes is absent in people with low concentrations of persistent organic pollutants in their blood. Furthermore, they find that the association between obesity and diabetes become stronger as

 $<sup>{}^{5}</sup>$ The British Medical Bulletin gathers several studies on the impact of environmental pollution on health in the issue 1 of the volume 68, in 2003

the concentrations of such pollutants in the blood increase. Lockwood (2002) give further references on the association between exposure to dioxins and the development of diabetes or altered insulin metabolism. He notes that obesity may increase the risk of diabetes: because insulin is concentrated in body fat, obese individuals are likely to have an increased dioxin body burden that could explain the link between the rise in the prevalence of diabetes and the "epidemic" of obesity (see also Ando et al., 2002, for empirical evidences).

One important feature of chronic diseases is that they do not affect only the oldest people. It impacts working-age people through illness and disability: according to the World Health Organization (WHO, 2004) 56% of those who support the burden of disease are people age 15-59 in high-income countries. Lakdawalla et al. (2004) demonstrate that the disability increases more rapidly for the young while it decreases for the elderly. They find that one responsible is the growth in asthma which appears to be enough to explain the change in disability. This is confirmed by Bhattacharya et al. (2008): "Recent work has shown that rates of severe disability, measured by the inability to perform basic activities of daily living, have been rising in working age populations. At the same time, the prevalence of important chronic diseases has been rising, while others falling, among working age populations. Chronically ill individuals are more likely than others to have activity of daily living limitations." Perlkowski and Berger (2004) study the influence of health on working conditions (wages and hours worked) making distinction between shortterm and long-term illness and the age at each illness appears (because it implies different adjustments for young people at early stage and for old people closed to retirement). They distinguish between temporary and permanent illness and find that the adverse effects of permanent health problems peak with an age of onset in the 40s for men and in the 30s for women. The biggest decline in wages and hours worked are observed for individuals whose problems started at those ages.

That leads to major economic impact in terms of labor productivity, labor supply, education or savings, as shown by Suhrcke et al. (2006a). Chronic diseases mainly conduce to a reduction of the productivity of the labor-force (even if agents are not sick enough to stop working or even if he is not sick but it is the member of her family who is sick) and the increase in disability of working people. For the US in 2003, Davis et al. (2005) estimated that 55 million workers over 148 million ages 19 to 64 reported the inability to concentrate at work because of their own illness or that of their family and 69 million workers reported missing days due to illness. About the effects of chronic diseases on workers' productivity, Blanc et al. (2001) demonstrate, with a sample of 125 adults in Northern-California that "Both asthma and rhinitis negatively affect work productivity. Those with asthma are less likely to be employed at all, while among those remaining on the job, rhinitis is a more potent cause of decreased work effectiveness. The economic impact of asthma and rhinitis and related conditions may be under-appreciated". For Australia, van Leeuwen et al. (2006) estimate that "while the impact of reduced work effectiveness on days worked with pain on productivity is uncertain, it has the potential to account for the majority of lost productivity costs associated with chronic pain.". Devol and Bedroussian (2007) from the Milken Institute estimate that the seven common chronic diseases represent for

the United-States lost productivity equals to \$1.1 trillion per year.

The second important features of chronic illness comes from that it places a burden on health care and welfare systems. For the United States in 2003, Devol and Bedroussian (2007) estimate that \$277 billion are spent annually on treatment. All these resources could be used in alternative activities promoting final (non health) consumption and growth.<sup>6</sup>

Furthermore, chronic illness has major implications in terms of occupational choices, that can not be supported (or funded) by public health-care system or insurance contracts. Since the Grossman's seminal work of 1972, time besides goods appears as an important input of the health production function by influencing the next period's health capital level. More recently, time and time costs in health production become important in the economic analysis of obesity for example (see Cutler et al., 2003; Philipson and Posner, 2008, for example). As emphasized by Mullahy and Robert (2008), increasing level of physical activities is now viewed as a mean to improve health outcomes. In their study based on the Bureau of Labor Satistics' American Time Use Survey Russell et al. (2007) noted that 11.3% of American adults (in 2003-2004) reported spending time (mean, 108 minutes) on activities related to health on their designed day and 5.6% (86 minutes) reported making medicine, giving self a shot, exercising or therapy for medical reasons. Physical activity has been shown to reduce the risk of developing or dying from heart diseases, diabetes, colon cancer and high blood-pressure. The US Department of Health and Human services gives some advice for being in good health to make each week 150 minutes of physical activities at moderate level or 75 minutes at vigourous level.

As a result, chronic illness force agents to allocate more time to health-enhancing activities, time they could use to home or market production. Therefore, both the increase in health-care expenditures and the rising investment in health-enhancing activities lead to a competition for resources that could be detrimental for economic activity, growth and/or welfare. And because pollution favours chronic illness and disability, this influence could be a new channel of transmission between the environmental public policy and the economy.

We investigate that point in the following sections.

# 3 The model

Let's consider an overlapping generations model. A new generation is born at each date t = 1, 2, ..., and lives for two periods. The number of individuals born at time t is L. Population is constant. Individuals are non-altruistic: the old do not care for the young and the young do not care for the old. The preferences of an agent born in period t are represented by the following utility (from van Zon and Muysken (2001)):

$$\log\left(c_{1t}^{\phi}h_t^{1-\phi}\right) + \theta\log\left(c_{2t+1}^{\phi}h_{t+1}^{1-\phi}\right)$$

<sup>&</sup>lt;sup>6</sup>Here, we are reasoning for a given life expectancy and we do not consider that life expectancy may rise or decrease. As a result, we do not integrate the fact that additional years of life increases health-care expenditures (see Suhrcke et al., 2008, p.15).

where  $c_{1t}$  and  $c_{2t+1}$  are respectively consumption in young and in old age,  $h_t$  and  $h_{t+1}$  are respectively private health-status in young and in old age. Parameter  $\theta = (1 + \iota)^{-1}$  where  $\iota > 0$  is the subjective discount rate of the agent. Parameters  $\phi > 0$  (respectively  $1 - \phi$ ) captures the relative importance of consumption (respectively health), in utility.<sup>7</sup>

Each young agent is endowed with one unit of time. She supplies  $\nu_t \in [0, 1]$  of this unit of time in final production and uses her remaining time  $1 - \nu_t$  as an investment in health care activities to improve her health status.<sup>8</sup> She earns a wage income  $\nu_t w_t$  where  $w_t$  is the wage rate.

The private health status of an agent born at period t evolves from period t and period t+1 according to two opposite forces (Aisa and Pueyo (2004)). On the one hand, biological processes involve a natural decay of health simply as time passes. Following Grossman (1972) and Cropper (1981) we further assume that health depreciates over time with the stock of pollution (denoted  $S_t$ ). On the other hand, the time invested in health-enhancing activities  $(1 - \nu_t)$  fights against this deterioration. Therefore, for an agent born at t, private health status evolves from period t to period t + 1 as:

$$h_{t+1} - h_t = \eta (1 - \nu_t) - \xi S_t^{\gamma} h_t \tag{1}$$

with  $\eta > 0$  is a productivity scalar of health-enhancing activities.<sup>9</sup>. Parameter  $\gamma \ge 0$  measures the influence of pollution stock on the natural decay  $\xi \in ]0,1[.^{10}]$ 

A consumer, born at t, works during the first period of her life, consumes an amount  $c_{1t}$  and saves the remainder of her revenue. The budget constraint of a young is

$$c_{1t} + s_t = \nu_t w_t$$

where  $s_t$  denotes saving in young. The budget constraint of an old is

$$c_{2t+1} = (1 + r_{t+1})s_t$$

where  $r_{t+1}$  is the interes rate paid on saving held from period t to t+1.

Firms operate through perfect competition using physical capital and labor to produce a final good with a constant return Cobb-Douglas technology:

$$Y_t = \tilde{A}_t K_t^{\alpha} N_t^{1-\alpha}, \qquad \alpha \in ]0,1[$$

<sup>9</sup>Note that here, we model a linear relationship between the health-enhancing activities and the evolution of health-status which could be not empirically relevant. As demonstrated by Skinner et al. (2001): "nearly 20 percent of total Medicare expenditures appears to provide no benefit in terms of survival, nor is it likely that this extra spending improves the quality of life". Our assumption is made for simplicity.

<sup>10</sup>We impose  $\gamma \geq 0$  to investigate the absence of a detrimental impact of pollution on health, that is  $\gamma = 0$ . Nevertheless, it is expected that  $\gamma > 1$ , that is the higher the stock of pollution, the higher the detrimental effect of pollution, even if there is no empirical evidence on such a linear relationship.

<sup>&</sup>lt;sup>7</sup>We do not integrate green preferences because we will assume in the following that health status is affected by pollution.

<sup>&</sup>lt;sup>8</sup>We could assume that there exists a sector that produces health care services with labor and therefore a part  $\nu$  of labor is allocated to manufacturing production and a part  $1 - \nu$  is allocated to health-care production. We would find the same qualitative results (see Appendix A). Consequently, what we call *investment in health-enhancing activities* could be viewed as health-care expenditures. Our modelling has the advantage to lead to a simpler exposition of the model and the results. In Appendix E, we demonstrate that results are not modified when leisure time is introduced.

where  $Y_t$  is the aggregate output,  $K_t$  is the aggregate productive capital and  $N_t$  is labor.  $\tilde{A}_t$  is a productive scalar, assumed as constant for the moment:  $\tilde{A}_t \equiv A$ . Capital depreciates fully in the production process.<sup>11</sup>

The stock of pollution S from period t to period t + 1 increases because of the net flow pollution in the current period t and decreases according to the rate of natural purification of pollutants  $\mu \in ]0, 1[$ . The net flow of pollution in period t is the ratio between pollution emissions in period t, denoted  $E_t$ , and the abatement activities funded by the government, denoted  $D_t$ . We assume, as conventional, that polluting emissions arise from final production such that

$$E_t = zY_t, \qquad z \in ]0,1[$$

Parameter z measures the polluting capacity of the technology. Consequently the stock of pollution in period t + 1 is defined as:

$$S_{t+1} = \left(\frac{zY_t}{D_t}\right)^{\chi} + (1-\mu)S_t \tag{2}$$

where  $\chi > 0$  is the exogenous elasticity of pollution stock with respect to the net flow of pollution E/D.

# 4 The competitive equilibrium

The representative agent born in period t maximizes her utility function taking wages, the interest rate and the stock of pollution as given. She chooses consumption at both ages  $(c_{1t}, c_{2t+1})$  and the proportion of time  $\nu_t$  she uses in production:

$$\max_{\{c_{1t}, c_{2t+1}, \nu_t\}} \log\left(c_{1t}^{\phi} h_t^{1-\phi}\right) + \theta \log\left(c_{2t+1}^{\phi} h_{t+1}^{1-\phi}\right)$$
$$s.t.\begin{cases} c_{1t} + s_t = \nu_t w_t\\ c_{2t+1} = (1+r_{t+1})s_t\\ h_{t+1} = \eta(1-\nu_t) + (1-\xi S_t^{\gamma}) h_t \end{cases}$$

The first-order condition gives saving:

$$s_t = \left(\frac{\theta}{1+\theta}\right)\nu_t w_t \tag{3}$$

and the allocation of time into production:

$$\nu_t = \frac{\phi(1+\theta)}{\eta(1-\phi)\theta} h_{t+1} \quad \in ]0,1[ \tag{4}$$

Because  $\nu_t < 1$ , the health status of the old  $h_{t+1}$  is bounded to  $\frac{\eta(1-\phi)\theta}{\phi(1+\theta)}$ .<sup>12</sup>

<sup>&</sup>lt;sup>11</sup>The production process is over the course of a generation. If the annual depreciation rate is 10% (which is empirically relevant), 96% of the capital stock is depreciated over the course of a 30 year generation. Therefore, we assume that capital is fully used up in the production process. Considering a positive depreciation rate would not change the qualitative results.

 $<sup>^{12}\</sup>mathrm{See}$  van Zon and Muysken (1997, p.5) for a justification of the health status boundary. .

We assume that abatement  $D_t$  is provided by the government as a public good and financed by an environmental tax  $\tau$  on the source of pollution  $Y_t$  such that the public budget is balanced at each date:  $D_t = \tau Y_t$ . The low of motion of the stock of pollution, given by equation (2) becomes:

$$S_{t+1} = \left(\frac{z}{\tau}\right)^{\chi} + (1-\mu)S_t$$

Firms maximize their profit  $\pi_t = (1 - \tau)Y_t - (1 + r_t)K_t - w_tN_t$  and the demand for capital and labor is

$$(1-\tau)\alpha Y_t/K_t = 1 + r_t$$
$$(1-\tau)(1-\alpha)Y_t/N_t = w_t$$

The good market clearing yields:

$$K_{t+1} = s_t L$$

and the labor market clearing equates labor demand  $N_t$  to labor supply  $\nu_t L$ 

$$N_t = \nu_t L$$

The competitive equilibrium may be summarized by the following relations:

$$K_{t+1} = (1-\alpha) \left(\frac{\theta}{1+\theta}\right) (1-\tau) A K_t^{\alpha} (\nu_t L)^{1-\alpha}$$
(5)

$$\nu_t = \frac{\phi(1+\theta)}{\eta(1-\phi)\theta} h_{t+1} \qquad \Rightarrow \qquad \nu_t = \left(\frac{(1-\phi)\theta}{\phi(1+\theta)} + 1\right)^{-1} \left(1 + (1-\xi S_t^{\gamma})\frac{h_t}{\eta}\right) \tag{6}$$

$$h_{t+1} = \eta (1 - \nu_t) + (1 - \xi S_t^{\gamma}) h_t \tag{7}$$

$$S_{t+1} = \left(\frac{z}{\tau}\right)^{\chi} + (1-\mu)S_t \tag{8}$$

The steady-state is defined here as an equilibrium where physical capital, private health-status, pollution stock and the allocation of labor in production are constant at  $K^*$ ,  $h^*$ ,  $S^*$  and  $\nu^*$  respectively, defined as:

$$S^{\star} = \mathcal{S}(\tau) \equiv \frac{(z/\tau)^{\chi}}{\mu} \tag{9}$$

$$h^{\star} = \mathcal{H}(\tau) \equiv \eta \left[ \frac{\phi(1+\theta)}{(1-\phi)\theta} + \xi \left( \frac{(z/\tau)^{\chi}}{\mu} \right)^{\gamma} \right]^{-1}$$
(10)

$$\nu^{\star} = \mathcal{V}(\tau) \equiv \phi \left[ \phi + \frac{\xi(1-\phi)\theta}{(1+\theta)} \left( \frac{(z/\tau)^{\chi}}{\mu} \right)^{\gamma} \right]^{-1}$$
(11)

Consequently, the health status and the allocation of time in production are positively affected by the environmental tax  $\tau$ .

From equations (5), (10) and (11), the steady-state value of the physical capital stock is:

$$K^{\star} = \mathcal{A} \ (1-\tau)^{\frac{1}{1-\alpha}} \mathcal{V}(\tau)$$

with  $\mathcal{A} \equiv \left( (1-\alpha) \left( \frac{\theta}{1+\theta} \right) A \right)^{1/(1-\alpha)} L$ . Because  $Y^{\star} = \frac{1+\theta}{\theta(1-\alpha)} (1-\tau)^{-1} K^{\star}$ ,<sup>13</sup> we obtain the steady-state value of final output as a function of the environmental taxation  $\tau$ :

$$Y^{\star} = \mathcal{A}_1 \ (1-\tau)^{\frac{\alpha}{1-\alpha}} \left( \mathcal{B} + \xi \left( \frac{z^{\chi}}{\mu} \right)^{\gamma} \tau^{-\chi\gamma} \right)^{-1}$$
  
with  $\mathcal{A}_1 \equiv \mathcal{B} \left( \frac{1+\theta}{\theta(1-\alpha)} \right) \mathcal{A}$  and  $\mathcal{B} \equiv \frac{\phi(1+\theta)}{(1-\phi)\theta}$ .

**Proposition 1.** When endogenous investment in private health-status and the detrimental impact of pollution on health are taken into account, the relationship between the steady-state output and the environmental taxation has an inverted-U shape.

Below (respectively above) an environmental tax-level denoted  $\hat{\tau}$  and defined as

$$\chi\gamma\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma}\hat{\tau}^{-1} - \frac{\alpha}{1-\alpha}\mathcal{B}\hat{\tau}^{\chi\gamma} - \left(\frac{\alpha}{1-\alpha} + \chi\gamma\right)\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma} = 0.$$
 (12)

a tighter environmental taxation rises (respectively lowers) the steady-state level of output  $Y^*$ .

#### **Proof**. See Appendix B

To understand the basic mechanism of this result, let us remember that

$$Y^{\star} = \mathcal{A}_1 \underbrace{(1-\tau)^{\frac{\alpha}{1-\alpha}}}_{Ia} \underbrace{\mathcal{V}(\tau)}_{Ib}$$
(13)

The environmental tax influences the steady-state level of output through two channels: the direct impact of the environmental taxation on the rewards to labor (see overbrace *Ia* in equation 13) and the (indirect) impact on the allocation of labor into the manufacturing sector (see overbrace *Ib* in equation 13).

The first one (negative) is the conventional "drag-down" effect of the environmental tax that reduces factor rewards – captured by  $(1-\tau)^{\frac{\alpha}{1-\alpha}}$ . The second one (positive) is a new channel of transmission due to the "competition for resources" between health enhancing activities and production activities that affects the supply of labor into the final production sector  $\nu^* = \mathcal{V}(\tau)$ . Because pollution has a detrimental impact on the evolution of private health-status, by reducing the net flow of pollution and therefore the stock of pollution, the environmental policy improves the private health-status of the agents. Consequently, each agent decides to reduce her investment in health enhancing activities  $(1 - \nu^*$  decreases)

<sup>13</sup>Note that 
$$K_{t+1} = s_t L = \left(\frac{\theta}{1+\theta}\right) \nu_t (1-\tau)(1-\alpha) Y_t / \nu_t = \left(\frac{\theta}{1+\theta}\right) (1-\tau)(1-\alpha) Y_t.$$

and to rise her labor supply to productive activities ( $\nu^*$  increases). In this way, the tighter environmental tax frees ressources that were allocated to health enhancing activities and that are now reallocated to production, leading to a higher level of steady-state output and a steady-state physical capital. Consequently the competition for resources between output production and health enhancing activities associated with the negative impact of pollution to health is a source of a new channel trough which the environmental policy may promote economic activity.

When  $\gamma = 0$ , the evolution of health-status is independent from pollution and therefore the investment of each agent in health-enhancing activities is not affected by the environmental tax:  $\nu^*$  is independent from  $\tau$ . In such a case, the competition for resources is not affected by the environmental policy and only the "drag-down" effect remains: the environmental policy is detrimental for growth. In the same way, the "competition for resources effect" does no longer hold when there is no endogenous investment in health-enhancing activities.

**Proposition 2.** The endogenous investment in health and the detrimental impact of pollution on health are two necessary conditions to obtain Proposition 1.

**Proof.** See above and Appendix B.

Considering the influences of parameters on the environmental tax-level  $\hat{\tau}$ , enables to understand why these two opposite effects of the environmental policy leads to an inversed-U shaped relationship between the environmental tax and the steady-state output level. These influences are summarized in the following table (see equation (12) in Proposition 1 and Appendix B for the demonstration):

ξ	$\eta$	$\phi$	$\theta$	$\alpha$	$\mu$	z
_	_	_	+	_	_	+

Table 1. Parameter Changes and Responses of  $\hat{\tau}$ 

Because it is cumbersome to obtain analytically the influence of  $\gamma$  and  $\chi$  on the taxlevel  $\hat{\tau}$ , we use a numerical application. We first calibrate the model assuming that the length of each period is 30 years, as usually in the literature. The first period covers ages 20 to 50, and the second period covers ages 50 to 80. We use the U.S. economy as benchmark. From De La Croix and Michel (2002), we choose  $\alpha$  and  $\theta$  following the standard choice in the RBC literature, that is  $\alpha = 0.36$  and a quarterly psychological discount factor equal to 0.99. It implies that  $\theta = 0.99^{(4\times30)} = 0.3$ . We use the calibration by van Zon and Muysken (1997) for the values of  $\xi$  and  $\phi$ . Finally, parameter  $\eta$  is chosen to obtain a private health-status higher than unity to enable the welfare to be positive.

Benchmark value of parameters are summarized in Table 2:

α	$\theta$	$\phi$	ξ	A	$\eta$	L	$\chi$	$\mu$	$\gamma$
0.36	0.3	1/2	0.2	50	0.8	1	1	0.5	1.5

Table 2. Benchmark value of parameters

	$\hat{ au}$	$\hat{Y}$	$\hat{\nu}^{\star}$	$\hat{h}^{\star}$	$\hat{S}^{\star}$	$\hat{W}^{\star}$
Benchmark	21.6%	9.73	0.896	2.068	1.84	1.28
$\gamma = 0.5$	7.77%	10.77	0.90	2.089	5.15	1.46
$\gamma = 1$	15.73%	10.12	0.894	2.065	2.54	1.35
$\gamma = 2$	25.97%	9.48	0.901	2.080	1.54	1.23
$\gamma = 2.5$	29.13%	9.31	0.907	2.094	1.37	1.20
$\chi = 1.25$	23.14%	9.769	0.909	2.099	1.667	1.28
$\chi = 1.5$	24.07%	9.821	0.921	2.125	1.515	1.29
$\chi = 1.75$	24.65%	9.875	0.930	2.146	1.387	1.293
$\chi = 2$	25.01%	9.928	0.937	2.163	1.279	1.298

and the results of the numerical application is reported in Table 3

Table 3. Steady-state  $\hat{\tau}$  for different values of  $\gamma$  and  $\chi$ 

The third proposition stems from the Table 1 and Table 3.

**Proposition 3.** When the negative impact of the environment on health and the endogenous decision of each agent to invest her resources into health enhancing activities are taken into account, we demonstrate that the environmental taxation will be more likely to improve the steady-state level of output if the rate of natural health decay ( $\xi$ ) is low, the efficiency of the health care spending ( $\eta$ ) is low, the weight of health in preferences  $(1 - \phi)$  is high, the part of labor in final output  $(1 - \alpha)$  is high, the rate of natural purification of pollutants ( $\mu$ ) is low, the polluting capacity of production technology (z) is high, the detrimental impact of pollution on health ( $\gamma$ ) and the elasticity of pollution stock with respect to the net flow of pollution  $\chi$  are high.

#### **Proof.** See Appendix B and Table 3.

When the detrimental effects of a dirty environment on health are important, the gains in terms of health status to reduce the emissions of pollutant are very important and the "competition for resources effect" that leads to an increase in labor supply runs beyond the "drag-down effect" that reduces factor rewards and as a consequence saving and physical capital accumulation. Neverthless, these positive gains diminish with the increase in the tax rate because the possible improvements in health-status due to the tax are reducing. In the same time, the losses from the reduction of factor rewards increase in the tax rate such that for the environmental tax-level  $\hat{\tau}$ , they offset the gains, and a further increase in  $\tau$  leads to a decrease in the steady-state output level.

Consequently, the greater the room for improving the environment and the private health-status through the environmental policy, the more beneficial the environmental policy is likely to be for the economy.

The numerical application also enables us to investigate the impact of the environmental taxation on the steady-state lifetime welfare, as well. As shown by Figure 1, there exists a relationship that is inverted-U shaped for similar reasons than the relationship between the steady-state output and the environmental tax is inverted-U shaped.



Figure 1: Steady-state lifetime welfare as a function of  $\tau$ 

# 5 Social optimum and the optimal environmental taxation

The purpose of this section is to investigate the determinants of the optimal environmental taxation in the presence of endogenous investment in private health and detrimental impact of pollution on private health.

In the centralized economy, the central planner aims at maximizing the welfare of agents:

$$\max_{\{c_1,c_2,\nu,K,D\}} \log\left(c_1^{\phi}h^{1-\phi}\right) + \theta \log\left(c_2^{\phi}h^{1-\phi}\right)$$
  
s.t. 
$$\begin{cases} Y = AK^{\alpha}(\nu L)^{1-\alpha} = Lc_1 + Lc_2 + D + K\\ h = \eta(1-\nu)/(\xi S^{\gamma})\\ S = (E/D)^{\chi}/\mu\\ E = zY \end{cases}$$

As demonstrated in Appendix C, consumption at young and old age are related<sup>14</sup>

$$\bar{c}_1 = \theta \bar{c}_2$$

with

$$\bar{c}_1 = \frac{(1-\alpha)\phi}{\alpha(1+\theta)} \left(\frac{A\alpha\phi}{(1-\alpha)\chi\gamma(1-\phi)+\phi}\right)^{1/(1-\alpha)}$$

The optimal allocation of time to production is:

$$\bar{\nu} = \phi$$

and the optimal stock of physical capital is

$$\bar{K} = \left(\frac{A\alpha\phi}{(1-\alpha)\chi\gamma(1-\phi)+\phi}\right)^{1/(1-\alpha)}\phi L$$

<sup>&</sup>lt;sup>14</sup>A bar <sup>–</sup> denotes optimal value.

Consequently, the optimal final output is

$$\bar{Y} = A^{1/(1-\alpha)} \left( \frac{\alpha \phi}{(1-\alpha)\chi\gamma(1-\phi) + \phi} \right)^{\alpha/(1-\alpha)} \phi L$$

Abatement activities is given by

$$\bar{D} = \frac{(1-\alpha)\chi\gamma(1-\phi)}{(1-\alpha)\chi\gamma(1-\phi)+\phi}\bar{Y}$$

and the optimal stock of pollution in the steady-state is

$$\bar{S} = \left(z\frac{(1-\alpha)\chi\gamma(1-\phi)+\phi}{(1-\alpha)\chi\gamma(1-\phi)}\right)^{\chi}/\mu$$

Consequently, the optimal value of the environmental tax that enables the decentralized economy to attain the optimal stock of pollution in the steady-state is

$$\bar{\tau} = \frac{(1-\alpha)\chi\gamma(1-\phi)}{(1-\alpha)\chi\gamma(1-\phi)+\phi}$$
(14)

It comes from this expression the following proposition.

**Proposition 4.** The higher the weight of health in preferences  $(1 - \phi)$ , the elasticity of pollution stock with respect to the net flow of pollution ( $\chi$ ), the detrimental impact of pollution on health (captured by  $\gamma$ ) and/or the part of labor in production  $(1 - \alpha)$ , the higher the optimal environmental tax is.

**Proof.** From equation (14), it is straightforward that  $\partial \bar{\tau} / \partial \phi < 0$ ,  $\partial \bar{\tau} / \partial \chi > 0$ ,  $\partial \bar{\tau} / \partial \gamma > 0$  and  $\partial \bar{\tau} / \partial \alpha < 0$ .

Nevertheless, the optimal environmental tax is not sufficient to enable the steadystate equilibrium to be optimal because in the decentralized economy the agents do not internalize the impact of their labor supply decisions on final output and the net flow of pollution. Consequently, agents supply not enough time to output production. That's why, to obtain the optimal individual labor supply  $\bar{\nu}$ , the government have to subsidy the health-enhancing activities at a rate (see Appendix D):

$$\bar{\tau^{\nu}} = 1 - \frac{\xi\theta}{(1+\theta)} \left(\frac{z^{\chi}}{\mu}\right)^{\gamma} \left(\frac{(1-\alpha)\chi\gamma(1-\phi)}{(1-\alpha)\chi\gamma(1-\phi)+\phi}\right)^{-\gamma\chi}$$

The environmental tax  $\bar{\tau}$  associated with the subsidy  $\bar{\tau^{\nu}}$  make the steady-state decentralized equilibrium optimal.

## 6 Extensions

#### 6.1 AK endogenous growth

In this section, we consider that there exists external learning by doing à la Romer (1986),<sup>15</sup> such that the productivity scalar  $\tilde{A}_t$  evolves as physical capital:

$$\tilde{A}_t = AK_t^{\alpha}$$

 $<sup>^{15}\</sup>mbox{Following Romer}$  (1986), production factors remain paid at their marginal after environmental tax cost.

to obtain an interest rate independent from physical capital.

Consequently, the final output becomes

$$Y_t = AK_t(\nu_t L)^{1-\alpha}$$

and the law of motion of physical capital is given by

$$K_{t+1} = (1-\alpha) \left(\frac{\theta}{1+\theta}\right) (1-\tau) A K_t(\nu_t L)^{1-\alpha}$$

At the steady-state, physical capital and output evolves at a constant positive rate of growth  $g^* \equiv K_{t+1}/K_t - 1$ , that is, using equation (11)

$$g^{\star} = \mathcal{A}'(1-\tau) \left[ \mathcal{B} + \xi \left( \frac{z^{\chi}}{\mu} \right)^{\gamma} \tau^{-\chi\gamma} \right]^{\alpha-1}$$

with  $\mathcal{A}' \equiv (1-\alpha) \left(\frac{\theta}{1+\theta}\right) A L^{1-\alpha} \mathcal{B}^{1-\alpha}$ . The influence of the environmental taxation on the growth rate at the steady-state is given by

$$\partial g^{\star} / \partial \tau = -\mathcal{A}' \left[ \mathcal{B} + \xi \left( \frac{(z/\tau)^{\chi}}{\mu} \right)^{\gamma} \right]^{\alpha - 1 - 1} \tau^{-\chi \gamma} \times \left[ \mathcal{B} + \xi \left( \frac{z}{\mu} \right)^{\gamma} \left( 1 + (1 - \alpha)\chi\gamma(1 - \tau^{-1}) \right) \right]$$

Consequently  $\partial g^* / \partial \tau > 0$  if the last term in the right-hand side of the previous expression is negative:

$$\mathcal{B}\tau^{\chi\gamma} + \xi \left(\frac{z}{\mu}\right)^{\gamma} \left(1 + (1-\alpha)\chi\gamma(1-\tau^{-1})\right) < 0$$

Because the left-hand side of the inequality is a monotonic increasing function of  $\tau$  with  $\lim_{\tau \to 0} = -\infty$  and  $\lim_{\tau \to 1} = \mathcal{B}\tau^{\chi\gamma} + \xi \left(\frac{z}{\mu}\right)^{\gamma} > 0$ , there exists a unique  $\hat{\tau}_g$  defined as

$$\mathcal{B}\hat{\tau}_g^{\chi\gamma} + \xi \left(\frac{z}{\mu}\right)^{\gamma} \left(1 + (1-\alpha)\chi\gamma(1-\hat{\tau}_g^{-1})\right) = 0$$

such that for  $\tau < \hat{\tau}_g$  (respectively  $\tau > \hat{\tau}_g$ ) we have  $\partial g^* / \partial \tau > 0$  (resp.  $\partial g^* / \partial \tau < 0$ ).

**Proposition 5.** Under the assumption of a learning-by-doing source of growth à la Romer (1986), the introduction of an endogenous private health care expenditures and a detrimental impact of pollution on health makes the environmental taxation policy good for growth when the level of taxation is not too high.

**Proof**. See above.

#### 6.2 Health affects labor productivity

As emphasized in the introduction, chronic diseases affect the economy through the huge losses of productivity they create. As a result, it is expected that a tighter environmental taxation will reduce these losses of productivity by reducing pollution and increasing the health-status of workers. To investigate how the "productivity effect" associated with the "competition for resources effect" could improve further the beneficial impact of the environmental policy, we introduce the impact of health on the productivity of labor.

We continue to consider that the poor health agents expend more on medical care when they are young and not elderly like Gutiérrez (2008). We do not assume that ill agents do not work (Williams (2002, 2003)). We rather consider that a better health-status makes workers more productive and that absenteeism due to illness does not occur.<sup>16</sup>

The technology to produce final output becomes:

$$Y_t = A_t K_t^{\alpha} (h_t^{\varepsilon} N_t)^{1-\alpha} \tag{15}$$

where  $\varepsilon \geq 0$  measures the effect of health on labor productivity. The introduction of health-dependent labor productivity lets the model unchanged except for the law of motion of physical capital (equation 5):

$$K_{t+1} = (1-\alpha) \left(\frac{\theta}{1+\theta}\right) (1-\tau) A K_t^{\alpha} (h_t^{\varepsilon} \nu_t L)^{1-\alpha}$$
(16)

As a result, the steady-state value of physical capital becomes:

$$K^{\star\star} = \mathcal{A} \ (1-\tau)^{\frac{1}{1-\alpha}} \mathcal{V}(\tau) \mathcal{H}(\tau)^{\varepsilon}$$

and the steady-state expression of final output is now:

$$Y^{\star\star} = \mathcal{A}_2 \ (1-\tau)^{\frac{\alpha}{1-\alpha}} \left( \mathcal{B} + \xi \left(\frac{z^{\chi}}{\mu}\right)^{\gamma} \tau^{-\chi\gamma} \right)^{-(1+\varepsilon)}$$

with  $\mathcal{A}_2 \equiv \mathcal{A}_1 \eta^{\varepsilon}$ .

**Proposition 6.** When the effect of health on labor productivity is taken into account, the positive effect of the environmental tax on ouput-level is enhanced and the tax level under which a thigher environmental tax increases output level is higher. The environmental policy is more likely to promote final output.

**Proof.** The tax level, denoted  $\hat{\tau}$ , for which  $\partial Y/\partial \tau = 0$  is

$$(1+\varepsilon)\chi\gamma\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma}\hat{\tau}^{-1} - \frac{\alpha}{1-\alpha}\mathcal{B}\hat{\tau}^{\chi\gamma} - \left(\frac{\alpha}{1-\alpha} + (1+\varepsilon)\chi\gamma\right)\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma} = 0.$$

It is straightforward that for  $\varepsilon = 0$ ,  $\hat{\tau} = \hat{\tau}$ . Furthermore the LHS of the equation is increasing in  $\varepsilon$  because  $\hat{\tau} \in ]0, 1[$ . Therefore from the theorem of implicit function we find that  $\partial \hat{\tau} / \partial \varepsilon > 0$ . Therefore  $\hat{\tau} > \hat{\tau}$  when  $\varepsilon > 0$ .

<sup>&</sup>lt;sup>16</sup>We take into account presenteeism, i.e. a worker present but with reduced productivity rather than absenteeism, i.e. a worker absent, because it accounts for not only worker health but also health of his family. For the US in 2003, Davis et al. (2005) estimated that 55 million workers over 148 million ages 19 to 64 reported the inability to concentrate at work because of their own illness or that of their family.

# 7 Conclusion

This paper investigated how the environmental tax affects the economy (output-level and -growth, welfare) when the detrimental impact of pollution on health is taken into account and working-age individuals have to invest in health-care activities to limit the deleterious influence of pollution.

In a two-period overlapping generations model, this paper demonstrates that the relationship between the environmental taxation and the economic activity (level- and growthoutput) becomes inverted-U shaped, when the detrimental impact of pollution on health and the private decision of each working-age agent to improve her health are taken into account. Especially, a tighter environmental tax is more likely to promote (rather than to harm) output-level and -growth when health is very sensitive to pollution, the weight of health in preferences is high, the polluting capacity of the production technology is high and the rate of natural purification of pollutants is low.

The inverted-U shaped relationship between the environmental tax and the economic activity is due to a positive effect arising from the competition for resources between the final output sector and the health-care sector that offsets the conventional detrimental "drag-down effect" for low values of the environmental tax.

We also demonstrate that the link between the environmental tax and the lifetime welfare is inverted-U shaped as well. Finally, we investigate the social optimum and the determinants of the optimal environmental tax.

This contribution shows that, besides life expectancy, there are other ways along which pollution may affect health and health affects economic activity. Those ways may be new channels of transmission of the environmental pollution to economic activity.

# References

- Agnani, B., Amaia, I., and Gutierrez, M. (2005). Growth in overlapping generation economies with non-renewable resources. *Journal of Environmental Economics and Management*, 50(2):300–318.
- Aisa, R. and Pueyo, F. (2004). Endogenous longevity, health and economic growth: a slow growth for a longer life? *Economics Bulletin*, 9(3):1–10.
- Ando, A., Yatagai, T., Rokkaku, K., Nagasaka, S., Ishikawa, S., and Ishibashi, S. (2002). Obesity is a critical risk factor for worsening of glucose tolerance in a family with the mutant insulin receptor. *Diabetes Care*, 25(8):1484–1485.
- Bhattacharya, J., Choudhry, K., and Lakdawalla, D. (2008). Chonic disease and severe disability among working-age populations. *Medical Care*, 46(1):92–100.
- Blanc, P., Trupin, L., Eisner, M., Earnest, G., Katz, P., Israel, L., and Yelin, E. H. (2001). The work impact of asthma and rhinitis: Findings from a population-based survey. *Journal of Clinical Epidemiology*, 54:610–618.

- Briggs, D. (2003). Environmental pollution and the global burden of disease. *British* Medical Bulletin, 68(1):1–24.
- Brook, R., Franklin, B., Cascio, W., Hong, Y., Howard, G., Lipsett, M., Luepker, R., Mittleman, M., Samet, M., Smith, S., and Tager, I. (2004). Air pollution and cardiovascular disease: a statement for healthcare professionials from the expert panel on population and prevention science of the american heart association. *Circulation*, 109:26655–2671.
- Chen, S. and Liao, C. (2005). Health risk assessment on humans exposed to environmental polycyclic aromatic hydrocarbons pollution sources. *Science of the Total Environment*.
- Cropper, M. (1981). Measuring the benefits from reduced morbidity. *The American Economic Review*, 71:235–240.
- Cutler, D., Glaeser, E., and Shapiro, J. (2003). Why have americans become more obese? Journal of Economic Perspectives, 17(3):93–118.
- Davis, K., Collins, S., Doty, M., Ho, A., and Holmgren, A. (2005). Health and productivity amoung u.s. workers. Issue brief, The Commonwealth Fund.
- De La Croix, D. and Michel, P. (2002). A Theory of Economic Growth : dynamics and policy in overlapping generations. Cambridge University Press.
- Devol, R. and Bedroussian, A. (2007). An unhealthy america: The economic burden of chronic disease – charting a new course to save lives and increase productivity and economic growth. Technical report, Milken Institute.
- Dormont, B., Grignon, M., and Huber, H. (2006). Health expenditure growth: Reassessing the threat of ageing. *Health Economics*, 15(9):947–963.
- Dormont, B., Martins, J., Pelgrin, F., and Suhrcke, M. (2007). Health expenditures, longevity and growth. Technical report.
- Grossman, M. (1972). On the concept of health capital and the demand for health. *Journal* of *Political Economy*, 80:223–255.
- Gutiérrez, M. (2008). Dynamic inefficiency in an overlapping generation economy with pollution and health costs. *Journal of Public Economic Theory*, 10(4):563–594.
- John, A. and Pecchenino, R. (1994). An overlapping generations model of growth and the environment. *The Economic Journal*, 104:1393–1410.
- John, A., Pecchenino, R., Schimmelpfenning, D., and Schreft, S. (1995). Short-lived agents and the long-lived environment. *Journal of Public Economics*, 58:127–141.
- Jones, O., MAguire, M., and Griffin, J. (2008). Environmental pllution and diabetes: a neglected association. *Lancet*, 371:287–288.
- Jouvet, P., Pestieau, P., and Ponthière, G. (2007). Longevity and environmental quality in an olg model. Working paper 2007-19, Université Paris X.

- Katsouyanni, K. (2003). Ambient air pollution and health. *British Medical Bulletin*, 68(1):143–156.
- Kemp, M. and Long, N. (1979). The under-exploitation of natural resources: A model with overlapping generations. *Economic Record*, 55:214–221.
- Lakdawalla, D., Bhattacharya, J., and Goldman, D. (2004). Are the young becoming mre disabled. *Health Affairs*, 23(1):168–176.
- Lang, A., Galloway, T., Scarlett, A., Henley, W., Depledge, M., Wallace, R., and Melzer, D. (2008). Association of urinary bisphenol a concentration with medical disorders and laboratory abnormalities in adults. *The Journal of the American Medical Association*, 300(11):1303–1310.
- Lee, D.-H., Lee, I.-K., Jin, S.-H., Steffes, M., and Jocbs, D. (2007). Association between serum concentrations of persistent organic pollutants and insulin resistance among nondiabetic adults: results from the national health and nutrition examination survey. *Diabetes Care*, 30:622–628.
- Lockwood, A. (2002). Diabetes and air pollution. Diabetes Care, 25(8):1487–1488.
- Mariani, F., Perez-Barahona, A., and Raffin, N. (2008). Life expectancy and the environment. Technical report, CES Working Paper 2008.48.
- Mourmouras, A. (1991). Competitive equilibria and sustainable growth in a life-cycle model with natural resources. *Scandinavian Journal of Economics*, 93(4):585–591.
- Mourmouras, A. (1993). Conservationist government policies and intergenerational equity in an overlapping generations model with renewable resources. *Journal of Public Economics*, 51(2):249–268.
- Mullahy, J. and Robert, S. (2008). No time to lose? time constraints and physical activity. Technical report, NBER Working Paper 14513.
- Nadal, M., Schuhmacher, M., and Domingo, J. (2004). Metal pollution of soils and vegetation in an area with petrochemical industry. *Science of the Total Environment*, 321:59–69.
- O'Neill, M., A.Veves, Zanobetti, A., Sarnat, J., Gold, D., Economides, P., Horton, E., and Schwartz, J. (2005). Diabetes enhances vulnerability to particulate air pollutionassociated impairment in vascular reactivity and endothelial function. *Circulation*, 111:2913–2920.
- Paulu, C., Aschengrau, A., and Ozonoff, D. (1999). Tetrachloroethylene-contaminated drinking water in Massachusetts and the risk of colon-rectum, lung, and other cancers. *Environmental Health Perspectives*, 107(4):265–271.

- Pautrel, X. (2008). Reconsidering the impact of pollution on long-run growth when pollution influences health and agents have a finite-lifetime. *Environmental and Resource Economics*, 40(1):37–52.
- Pautrel, X. (2009). Pollution and life expectancy: How environmental policy can promote growth. *Ecological Economics*, Forthcoming.
- Perlkowski, J. and Berger, M. (2004). The impact of health on employment, wages, and hours worked over life-cycle. The Quarterly Review of Economics and Finance, 44:102– 121.
- Philipson, T. and Posner, R. (2008). Is the obsety epidemic a public health problem? a decade of research on the economics of obesity. Technical report, NBER Working Paper 14010.
- Pope, C. and al. (2004). cardiovascular mortality and long-term exposure to particulate air pollution: epidemiological evidence of general pathological pathways of disease. *Circulation*, 109(1):71–77.
- Pope, C., Burnett, R., Thun, M., Calle, E., Krewski, D., Ito, K., and Thurston, G. (2002). Lung cancer, cardiopulmonary mortality, and long-term exposure to fine particulate air pollution. *The Journal of American Medical Association*, 287:1132–1141.
- Porta, M. (2006). Persistent organic pollutants and te burden of diabetes. *Lancet*, 368:558–559.
- Rajagopalan, S., Sun, Q., and Chen, L. C. (2005). Particulate pollution and endothelial function: Déjà vu all over again in the air. *Circulation*, 111(2869-2871).
- Remillard, R. and Bunce, N. (2002). Linking dioxins to diabetes: epidemiology and biological plausibility. *Environmental Health Perspectives*, 110:853–858.
- Romer, P. (1986). Increasing returns and long run growth. *Journal of Political Economy*, 94:1002–1037.
- Russell, L., Ibuka, Y., and Abraham, K. (2007). Health-related activities in the american time use survey. *Medical Care*, 45(7):680–685.
- Rylander, L., Rignell-Hydbom, A., and Hagmar, L. (2005). A cross-sectional study of the association between persistent organochlorine pollutants and diabetes. *Environmental Health*, 4:28.
- Schuhmacher, M. and Domingo, J. (2006). Long-term study of environmental levels of dioxins and furans in the vicinity of a municipal solid waste incinerator. *Environment International*, 32:397–404.
- Skinner, S., Fisher, E., and Wenberg, J. (2001). The efficiency of medicare. National Bureau of Economic Research Working paper 8395, National Bureau of Economic Research.

- Smink, A., Ribas-Fito, N., Garcia, R., Torrent, M., Mendez, M., Grimalt, J., and Sunyer, J. (2008). Exposure ti hexachlorobenzene during pregnancy increases the risk of overweight in children aged 6 years. *Acta Pediatrica*, 97:1465–1469.
- Suhrcke, M., Arce, R., Mckee, M., and Rocco, L. (2008). Economic costs of ill healthin the european region. Background document, World Health Organization.
- Suhrcke, M., Mckee, M., Stuckler, D., Arce, R., Tsolova, S., and Mortensen, J. (2006a). The contribution of health to the economy in the european union. *Public Health*, 120:994–1001.
- Suhrcke, M., Nugent, R., Stuckler, D., and Rocco, L. (2006b). Chronic disease: An economic perspective. Technical report, Oxford Health Alliance, London.
- Valent, F., Little, D., Bertollini, R., Nemer, L., Barbone, F., and Tamburlini, G. (2004). Burden of disease attributable to selected environmental factors and injury among children and adolescents in Europe. *The Lancet*, 363:2032–2039.
- van Leeuwen, M., Blyth, F., March, L., Nicholas, M., and Cousins, M. J. (2006). Chronic pain and reduced work effectiveness: The hidden cost to australian employers. *European Journal of Pain*, 10:161–166.
- van Zon, A. and Muysken, J. (1997). Health, education and endogenous growth. Technical Report 97-009, MERIT Research Memorandum.
- van Zon, A. and Muysken, J. (2001). Health and endogenous growth. Journal of Health Economics, 20:169–185.
- Varvarigos, D. (2008). Environmental quality, life expectancy and sustainable economic growth. Working paper 08/19, University of Leicester, Departement of Economics.
- WHO (2004). The global burden of disease: 2004 update. Technical report, World Health Organization.
- WHO (2005). Preventing chronic diseases: A vital investment (who, geneva, 2005). Technical report, World Health Organization, Geneva.
- Williams, R. (2002). Environmental tax interactions when pollution affects health or productivity. Journal of Environmental Economics and Management, 44(2):261–270.
- Williams, R. (2003). Health effects and optimal environmental taxes. Journal of Public Economics, 87:323–335.
- Zhang, X., Zhao, X., and Harris, A. (2008). Chronic diseases and labour force participation in australia. *Journal of Health Economics*.

# Appendix

## A The basic model with a health sector

Let us consider in this section, that there exists in the economy a health sector that produces an amount  $H_t$  of health-care services at period t, using labour and the following technology

$$H_t = A_H (1 - \nu_t)$$

where  $A_H$  is a productivity scalar and  $1 - \nu_t$  is the part of labour (normalized to unity) allocated to the health sector.

Firms in the health sector operate under perfect competition and maximize their profit  $m_t H_t = w_t (1 - \nu_t)$  such that:

$$m_t A_H = w_t \tag{17}$$

The final output sector is always defined in section 2.

In the competitive equilibrium, besides her consumption of final good, the consumer buys when she is young  $H_t$  units of health care services for an amount of health-care expenditures equal to  $m_t H_t$ . The private health status of the agent evolves between period t and t + 1 as:

$$h_{t+1} - h_t = \eta H_t - \xi S_t^{\gamma} h_t$$

The program of the consumer consists in choosing consumption when young and old and health-care services  $H_t$  in order to maximize her lifetime utility subject to her budget constraint when young and old and the evolution of her health-status:

$$\max_{\{c_{1t}, c_{2t+1}, H_t\}} \log \left( c_{1t}^{\phi} h_t^{1-\phi} \right) + \theta \log \left( c_{2t+1}^{\phi} h_{t+1}^{1-\phi} \right)$$
$$s.t. \begin{cases} c_{1t} + m_t H_t + s_t = w_t \\ c_{2t+1} = (1+r_{t+1})s_t \\ h_{t+1} = \eta H_t + (1-\xi S_t^{\gamma}) h_t \end{cases}$$

The first-order condition gives saving (using equation 17):

$$s_t = \left(\frac{\theta}{1+\theta}\right) (w_t - m_t H_t) = \left(\frac{\theta}{1+\theta}\right) \nu_t w_t$$

Furthermore,  $\nu_t$  is given by

$$h_{t+1} = \frac{\eta A_H (1-\phi)\theta}{\phi(1+\theta)} \nu_t$$

that is  $\nu_t$  increases in  $h_{t+1}$ . These two expressions are similar to those find with health-care investment as time (see equations (3) and (4)) when  $A_H = 1$ .

## **B** Environmental taxation in the competitive equilibrium

The influence of the environmental tax on the steady-state level of output is given by:

$$dY^{\star}/d\tau = Y^{\star}\tau^{-\gamma\chi}(1-\tau)^{-1}\left(\mathcal{B} + \xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma}\tau^{-\chi\gamma}\right)^{-1} \times \left[\chi\gamma\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma}\tau^{-1} - \frac{\alpha}{1-\alpha}\mathcal{B}\tau^{\chi\gamma} - \left(\frac{\alpha}{1-\alpha} + \chi\gamma\right)\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma}\right].$$

The influence of the environmental tax on the steady-state level of output is positive if and only if

$$\chi\gamma\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma}\tau^{-1} - \frac{\alpha}{1-\alpha}\mathcal{B}\tau^{\chi\gamma} - \left(\frac{\alpha}{1-\alpha} + \chi\gamma\right)\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma} > 0$$

Because the left-hand side of the inequality is a decreasing monotonic function of  $\tau$  with  $\lim_{\tau \to 0} = +\infty$  and  $\lim_{\tau \to 1} = \frac{-\alpha}{1-\alpha} \left( \mathcal{B} + \xi \left( \frac{z^{\chi}}{\mu} \right)^{\gamma} \right) < 0$ , there exists a unique  $\tau \in ]0, 1]$  under which the inequality is verified. This  $\tau$  is denoted  $\hat{\tau}$  and is defined as:

$$\chi\gamma\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma}\hat{\tau}^{-1} - \frac{\alpha}{1-\alpha}\mathcal{B}\hat{\tau}^{\chi\gamma} - \left(\frac{\alpha}{1-\alpha} + \chi\gamma\right)\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma} = 0.$$

When  $\gamma = 0$ , the left-hand side of the inequality is independent from  $\tau$  and negative. Consequently, when  $\gamma = 0$ , we have  $dY^*/d\tau < 0$ .

To find how parameters affect the tax level  $\tau$ , let rewrite the expression of  $\hat{\tau}$  as:

$$\Gamma(\hat{\tau};\alpha,\gamma,\xi,\mu,\mathcal{B},z,\chi) \equiv \mathcal{B}\hat{\tau}^{\chi\gamma} + \left(1 - \alpha^{-1}(1-\alpha)\chi\left(\hat{\tau}^{-1} - 1\right)\gamma\right)\xi\left(\frac{z^{\chi}}{\mu}\right)^{\gamma} = 0.$$

with  $\mathcal{B} \equiv \eta \frac{\phi(1+\theta)}{(1-\phi)\theta}$  and  $\hat{\tau}^{-1} > 1$ . Except for  $\gamma$  and  $\chi$ , it is straightforward that  $\partial \Gamma(\cdot)/\partial \xi > 0$ ,  $\partial \Gamma(\cdot)/\partial \mathcal{B} > 0$ ,  $\partial \Gamma(\cdot)/\partial \alpha > 0$ ,  $\partial \Gamma(\cdot)/\partial \mu > 0$ ,  $\partial \Gamma(\cdot)/\partial z < 0$ ,  $\partial \Gamma(\cdot)/\partial \hat{\tau} > 0$ . From the theorem of implicit function, we obtain

$$\begin{split} \partial \hat{\tau} / \partial \xi < 0, \qquad \partial \hat{\tau} / \partial \eta < 0, \qquad \partial \hat{\tau} / \partial \phi < 0, \qquad \partial \hat{\tau} / \partial (1-\phi) > 0, \qquad \partial \hat{\tau} / \partial \theta > 0, \\ \partial \hat{\tau} / \partial \alpha < 0, \qquad \partial \hat{\tau} / \partial \mu < 0, \qquad \partial \hat{\tau} / \partial z > 0. \end{split}$$

# C The optimum

In the centralized economy, the central planner aims at maximizing the welfare of agents:

$$\max_{\{c_1,c_2,\nu,K,D\}} \log\left(c_1^{\phi}h^{1-\phi}\right) + \theta \log\left(c_2^{\phi}h^{1-\phi}\right)$$
s.t.
$$\begin{cases} F(K,\nu,L) = \tilde{A}K^{\alpha}(\nu L)^{1-\alpha} = Lc_1 + Lc_2 + D + K\\ h = \eta(1-\nu)/(\xi S^{\gamma})\\ S = (E/D)^{\chi}/\mu\\ E = zY \end{cases}$$

The Lagrangian may be written as:

$$\mathcal{L} = \phi \left( \log c_1 + \theta \log c_2 \right) + (1+\theta)(1-\phi) \left( \chi \gamma \log D - \chi \gamma \log F(K,\nu,L) + \log(1-\nu) \right) + (1+\theta)(1-\phi) \log \frac{\eta \mu^{\gamma}}{\xi z^{\gamma \chi}} + \lambda \left( F(K,\nu,L) - Lc_1 - Lc_2 - D - K \right)$$

First-order conditions are

$$\phi c_1^{-1} = \lambda L \tag{18}$$
$$\theta \phi c_2^{-1} = \lambda L$$

that is

$$c_2 = \theta c_1$$

$$\lambda(F'_K(\cdot) - 1) = (1 + \theta)\chi\gamma(1 - \phi)F'_K(\cdot)/F(\cdot)$$
(19)

$$(1-\phi)(1+\theta)\left(\gamma\chi F_{\nu}'(\cdot)/F(\cdot) + (1-\nu)^{-1}\right) = \lambda F_{\nu}'(\cdot)$$
(20)

$$\lambda = (1+\theta)\chi\gamma(1-\phi)D^{-1} \tag{21}$$

Equations (19) and (21) give:

$$D = Y(1 - 1/F'_{K}(\cdot)) = Y - \alpha^{-1}K$$

Furthermore, from (18), we obtain  $\lambda = \frac{\phi}{c_1 L}$ , consequently

$$D = \phi^{-1}(1+\theta)\chi\gamma(1-\phi)c_1L$$

and consequently the market equilibrium gives

$$Y = c_1 L(1+\theta) \left[ 1 + \phi^{-1} \chi \gamma (1-\phi) \right] + K$$

that is

$$c_1 L = \frac{Y - K}{(1 + \theta) \left[1 + \phi^{-1} \chi \gamma (1 - \phi)\right]}$$

In the same way, the market equilibrium may be written as

$$Y = (1 + \theta)c_1 L + K + Y - \alpha^{-1} K$$

that is

$$(1+\theta)c_1L = (\alpha^{-1} - 1)K$$

Consequently

$$\frac{Y - K}{[1 + \phi^{-1}\chi\gamma(1 - \phi)]} = (\alpha^{-1} - 1)K$$

that is

$$Y = \alpha^{-1} \phi^{-1} \left[ (1 - \alpha) \chi \gamma (1 - \phi) + \phi \right] K$$
(22)

Finally, equation (20) gives

$$(1-\phi)(1+\theta)\left((1-\alpha)\gamma\chi + \frac{\nu}{1-\nu}\right) = \frac{\phi(1+\theta)}{(\alpha^{-1}-1)}(1-\alpha)Y/K$$

that is

$$\frac{\nu}{1-\nu} = \frac{\phi}{1-\phi} \qquad \Rightarrow \qquad \nu = \phi$$

From (22), we can obtain the express of the steady-state physical capital in the centralized economy, denoted  $K_c$ :

$$\bar{K} = \left(\frac{A\alpha\phi}{(1-\alpha)\chi\gamma(1-\phi)+\phi}\right)^{1/(1-\alpha)}\phi L$$

and from (22)

$$\bar{Y} = A^{1/(1-\alpha)} \left( \frac{\alpha \phi}{(1-\alpha)\chi\gamma(1-\phi) + \phi} \right)^{\alpha/(1-\alpha)} \phi L$$

Finally

$$\bar{c}_1 = \frac{(1-\alpha)}{\alpha(1+\theta)} \left(\frac{A\alpha\phi}{(1-\alpha)\chi\gamma(1-\phi)+\phi}\right)^{1/(1-\alpha)}\phi$$

and

$$\bar{D} = \frac{(1-\alpha)\chi\gamma(1-\phi)}{(1-\alpha)\chi\gamma(1-\phi) + \phi}\bar{Y}$$

# D Subsidy to health-enhancing activities

We re-write the competitive equilibrium assuming that the government subsidies the health-enhancing activities by agent by paying a subsidy  $\tau^{\nu}$  to the opportunity cost of health-enhancing activities (that is the foregone wage  $(1 - \nu_t)w_t$ ) that is funded by a lump-sum tax denoted  $a_t$ .

The budget-constraint for the young born at period t becomes:

$$c_1 + s_t + a_t = \nu_t w_t + \tau^{\nu} (1 - \nu_t) w_t$$

The maximization of lifetime utility gives  $s_t = \frac{\theta}{1+\theta}(\nu_t w_t + \tau^{\nu}(1-\nu_t)w_t - a_t)$  and because government budget constraint requires  $a_t = \tau^{\nu}(1-\nu_t)w_t$ , we obtain

$$s_t = \frac{\theta}{1+\theta} \nu_t w_t$$

and the individual labor supply is given by

$$\nu_t = \frac{(1+\theta)\phi}{\theta(1-\phi)} (1-\tau^{\nu}) h_{t+1}$$

In the steady-state equilibrium, the private health-status remains constant at:

$$h^{\star} = \mathcal{H}(\tau) \equiv \eta \left[ (1 - \tau^{\nu}) \frac{\phi(1 + \theta)}{(1 - \phi)\theta} + \xi \left( \frac{(z/\tau)^{\chi}}{\mu} \right)^{\gamma} \right]^{-1}$$
$$\nu^{\star} = \mathcal{V}(\tau) \equiv \phi \left[ \phi + \frac{\xi(1 - \phi)\theta}{(1 - \tau^{\nu})(1 + \theta)} \left( \frac{(z/\tau)^{\chi}}{\mu} \right)^{\gamma} \right]^{-1}$$

The higher the subsidy to health-enhancing activities, the higher the health-status in the steady-state and the lower the individual supply of labor  $\nu^*$ .

The subsidy to health-enhancing activities that enables to replicate the optimal allocation of time between health-enhancing activities and production (denoted  $\tau^{\bar{\nu}}$ ) is such that  $\nu^* = \phi$ , that is

$$\bar{\tau^{\nu}} = 1 - \frac{\xi\theta}{(1+\theta)} \left(\frac{z^{\chi}}{\mu}\right)^{\gamma} \left(\frac{(1-\alpha)\chi\gamma(1-\phi)}{(1-\alpha)\chi\gamma(1-\phi)+\phi}\right)^{-\gamma\chi}$$

# E The competitive equilibrium with leisure

Until here, we investigate the competition for resources assuming that the time unit at the disposal of the agent is divided between two different "occupational" activities: production which enables to earn an income and investment in health care which enables the agent to stay in good health.

In this section, we consider that each agent also values leisure-time and that he may adjust his leisure-time according to the level of his health-status: the healthier is the agent, the greater is the utility of one minute of leisure.

To do so, we continue to denote  $\nu \in ]0, 1[$ , the part time the agent chooses in output production and we denote  $u \in ]0, 1[$  the part time spent as investment in health status. Consequently  $1 - u - \nu \in ]0, 1[$  represents the part-time

To keep things simple, preferences are written as follows

$$\log\left(c_{1t}^{\phi}h_t^{1-\phi}(1-u_t-\nu_t)^{\phi_1}\right) + \theta\log\left(c_{2t+1}^{\phi}h_{t+1}^{1-\phi}\right)$$

with  $\phi_1 > 0$  captures the weight of leisure in utility.

The law of motion of the private health-status (equation 1) is modified as follows

 $h_{t+1} - h_t = \eta u_t - \xi S_t^{\gamma} h_t$ 

and in the competitive equilibrium, the program of the agent is

$$\max_{\{c_{1t}, c_{2t+1}, \nu_t, u_t\}} \log\left(c_{1t}^{\phi} h_t^{1-\phi} (1-u_t-\nu_t)^{\phi_1}\right) + \theta \log\left(c_{2t+1}^{\phi} h_{t+1}^{1-\phi}\right)$$
$$s.t. \begin{cases} c_{1t} + s_t = \nu_t w_t \\ c_{2t+1} = (1+r_{t+1})s_t \\ h_{t+1} = \eta u_t + (1-\xi S_t^{\gamma}) h_t \end{cases}$$

The expression of savings  $s_t$  is the same than equation (3) and the part-time into output production  $\nu_t$  is still given by equation (4). But now, the part-time in health care expenditures:

$$u_t = 1 - \frac{\phi(1+\theta)}{\eta(1-\phi)\theta} h_{t+1} - \frac{\phi_1}{\eta(1-\phi)\theta} h_{t+1} = 1 - \frac{\phi(1+\theta) + \phi_1}{\eta(1-\phi)\theta} h_{t+1}$$

and the part-time to leisure is

$$1 - u_t - \nu_t = \frac{\phi_1}{\eta(1 - \phi)\theta} h_{t+1}$$

The competitive equilibrium may be summarized by the following relations:

$$K_{t+1} = (1 - \alpha) \left(\frac{\theta}{1 + \theta}\right) (1 - \tau) A K_t^{\alpha} (\nu_t L)^{1 - \alpha}$$
$$\nu_t = \frac{\phi(1 + \theta)}{\eta(1 - \phi)\theta} h_{t+1}$$
$$h_{t+1} = \eta u_t + (1 - \xi S_t^{\gamma}) h_t$$
$$u_t = 1 - \frac{\phi(1 + \theta) + \phi_1}{\eta(1 - \phi)\theta} h_{t+1}$$
$$S_{t+1} = \left(\frac{z}{\tau}\right)^{\chi} + (1 - \mu) S_t$$

In the steady-state,  $\nu$ , u, h, S, K, Y remain constant. Consequently, the stock of pollution in the steady-state  $S^*$  is always given by (9), and the private health-status is

$$h^{\star} = \eta \left[ \mathcal{B}' + \xi \left( \frac{z^{\chi}}{\mu} \right)^{\gamma} \tau^{-\chi\gamma} \right]^{-1}$$

with  $\mathcal{B}' \equiv \frac{\phi(1+\theta) + \phi_1}{(1-\phi)\theta}$ . When leisure is taken into account, the health care expenditures is lowered and the steady-state health status too, but the influence of the environmental tax is not modified.

The part-time into the output production is:

$$\nu^{\star} = \mathcal{B} \left[ \mathcal{B}' + \xi \left( \frac{z^{\chi}}{\mu} \right)^{\gamma} \tau^{-\chi\gamma} \right]^{-1}$$

Finally, we obtain

$$Y^{\star} = \mathcal{A}_{1}^{\prime} \ (1-\tau)^{\frac{\alpha}{1-\alpha}} \left( \mathcal{B} + \xi \left( \frac{z^{\chi}}{\mu} \right)^{\gamma} \tau^{-\chi\gamma} \right)^{-(1+\varepsilon)}$$

with  $\mathcal{A}'_1 \equiv \mathcal{B}'\left(\frac{1+\theta}{\theta(1-\alpha)}\right)\mathcal{A}.$ 

Because  $\mathcal{A}'_1 < \mathcal{A}_1$ , the steady-state level of output is lower when leisure is taken into account, but the effect of the environmental taxation in the steady-state level of output and the expression of  $\tau$ , the environmental taxation under which a higher tax promotes economic activities, are not modified.

#### NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

#### Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses: http://www.feem.it/Feem/Pub/Publications/WPapers/default.htm http://www.ssrn.com/link/feem.html

http://www.repec.org

http://agecon.lib.umn.edu

http://www.bepress.com/feem/

#### NOTE DI LAVORO PUBLISHED IN 2009

SD	1.2009	Michael Hoel: <u>Bush Meets Hotelling: Effects of Improved Renewable Energy Technology on Greenhouse Gas</u>
		Emissions
SD	2.2009	Abay Mulatu, Reyer Gerlagh, Dan Rigby and Ada Wossink: <u>Environmental Regulation and Industry Location</u>
SD	3.2009	Anna Alberini, Stefania Tonin and Margherita Turvani: <u>Rates of Time Preferences for Saving Lives in the</u> Hazardous Waste Site Context
SD	4.2009	Elena Ojea, Paulo A.L.D. Nunes and Maria Loureiro: <u>Mapping of Forest Biodiversity Values: A Plural</u> Perspective
SD	5.2009	Xavier Pautrel : Macroeconomic Implications of Demography for the Environment: A Life-Cycle Perspective
IM	6 2009	Andrew Ellul Marco Pagano and Fausto Panunzi: Inheritance Law and Investment in Family Firms
IM	7 2009	Luigi Zingales: The Future of Securities Regulation
SD	8.2009	Carlo Carraro, Emanuele Massetti and Lea Nicita: How Does Climate Policy Affect Technical Change? An
		Analysis of the Direction and Pace of Technical Progress in a Climate-Economy Model
SD	9.2009	William K. Jaeger: <u>The Welfare Effects of Environmental Taxation</u>
SD	10.2009	Aude Pommeret and Fabien Prieur: Double Irreversibility and Environmental Policy Design
SD	11.2009	Massimiliano Mazzanti and Anna Montini: Regional and Sector Environmental Efficiency Empirical Evidence
		from Structural Shift-share Analysis of NAMEA data
SD	12.2009	A. Chiabai, C. M. Travisi, H. Ding, A. Markandya and P.A.L.D Nunes: Economic Valuation of Forest
		Ecosystem Services: Methodology and Monetary Estimates
SD	13.2009	Andrea Bigano, Mariaester Cassinelli, Fabio Sferra, Lisa Guarrera, Sohbet Karbuz, Manfred Hafner, Anil
		Markandya and Ståle Navrud: <u>The External Cost of European Crude Oil Imports</u>
SD	14.2009	Valentina Bosetti, Carlo Carraro, Romain Duval, Alessandra Sgobbi and Massimo Tavoni: <u>The Role of R&amp;D</u>
		and Technology Diffusion in Climate Change Mitigation: New Perspectives Using the Witch Model
IM	15.2009	Andrea Beltratti, Marianna Caccavaio and Bernardo Bortolotti: Stock Prices in a Speculative Market: The
		Chinese Split-Share Reform
GC	16.2009	Angelo Antoci, Fabio Sabatini and Mauro Sodini: <u>The Fragility of Social Capital</u>
SD	17.2009	Alexander Golub, Sabine Fuss, Jana Szolgayova and Michael Obersteiner: Effects of Low-cost Offsets on
<b>CD</b>	40.0000	Energy Investment – New Perspectives on REDG
SD	18.2009	Enrica De Cian: <u>Factor-Augmenting Technical Change: An Empirical Assessment</u>
SD	19.2009	Irene Valsecchi: Non-Uniqueness of Equilibria in One-Shot Games of Strategic Communication
SD	20.2009	Dimitra Vouvaki and Anastasios Xeapapadeas: <u>Total Factor Productivity Growth when Factors of Production</u>
SD	21 2009	Giulia Macanno, Maria Loureiro, Paulo A L D, Nunes and Richard Tol: Assessing the Impact of Biodiversity
50	21.2009	on Tourism Flows: A model for Tourist Behaviour and its Policy Implications
IM	22.2009	Bernardo Bortolotti, Veliko Fotak, William Megginson and William Miracky: Sovereign Wealth Fund
		Investment Patterns and Performance
IM	23.2009	Cesare Dosi and Michele Moretto: Auctioning Monopoly Franchises: Award Criteria and Service Launch
		Requirements
SD	24.2009	Andrea Bastianin: Modelling Asymmetric Dependence Using Copula Functions: An application to Value-at-
		Risk in the Energy Sector
IM	25.2009	Shai Bernstein, Josh Lerner and Antoinette Schoar: The Investment Strategies of Sovereign Wealth Funds
SD	26.2009	Marc Germain, Henry Tulkens and Alphonse Magnus: Dynamic Core-Theoretic Cooperation in a Two-
		Dimensional International Environmental Model
IM	27.2009	Frank Partnoy: <u>Overdependence on Credit Ratings Was a Primary Cause of the Crisis</u>
SD	28.2009	Frank H. Page Jr and Myrna H. Wooders (lxxxv): <u>Endogenous Network Dynamics</u>
SD	29.2009	Caterina Calsamiglia, Guillaume Haeringer and Flip Klijnb (lxxxv): Constrained School Choice: An
		Experimental Study
SD	30.2009	Gilles Grandjean, Ana Mauleon and Vincent Vannetelbosch (lxxxv): <u>Connections Among Farsighted Agents</u>
SD	31.2009	Antonio Nicoló and Carmelo Rodríguez Álvarez (lxxxv): Feasibility Constraints and Protective Behavior in
		Efficient Kidney Exchange
SD	32.2009	Rahmi İlkiliç (lxxxv): <u>Cournot Competition on a Network of Markets and Firms</u>
SD	33.2009	Luca Dall'Asta, Paolo Pin and Abolfazl Ramezanpour (lxxxv): Optimal Equilibria of the Best Shot Game
SD	34.2009	Edoardo Gallo (lxxxv): <u>Small World Networks with Segregation Patterns and Brokers</u>
SD	35.2009	Benjamin Golub and Matthew O. Jackson (lxxxv): How Homophily Affects Learning and Diffusion in
		Networks
SD	36.2009	Markus Kinateder (lxxxv): <u>Team Formation in a Network</u>
SD	37.2009	Constanza Fosco and Friederike Mengel (lxxxv): Cooperation through Imitation and Exclusion in Networks
SD	38.2009	Berno Buechel and Tim Hellmann (lxxxv): Under-connected and Over-connected Networks

SD	39.2009	Alexey Kushnir (lxxxv): <u>Matching Markets with Signals</u>
SD	40.2009	Alessandro Tavoni (lxxxv): Incorporating Fairness Motives into the Impulse Balance Equilibrium and Quantal
		Response Equilibrium Concepts: An Application to 2x2 Games
SD	41.2009	Steven J. Brams and D. Marc Kilgour (lxxxv): Kingmakers and Leaders in Coalition Formation
SD	42.2009	Dotan Persitz (lxxxv): Power in the Heterogeneous Connections Model: The Emergence of Core-Periphery
		<u>Networks</u>
SD	43.2009	Fabio Eboli, Ramiro Parrado, Roberto Roson: <u>Climate Change Feedback on Economic Growth: Explorations</u>
66	44.0000	with a Dynamic General Equilibrium Mode
GC	44.2009	Fabio Sabatini: Does Social Capital Create Trust? Evidence from a Community of Entrepreneurs
SD	45.2009	ZhongXiang Zhang: Is it Fair to Treat China as a Christmas Tree to Hang Everybody's Complaints? Putting
SD.	46 2000	Its Own Energy Saving into Perspective
50	40.2009	Efficition S. Sartzetakis, Anastasios Aepapadeas and Emmanuel Petrakis: <u>The Role of Information Provision</u>
SD	47 2009	lean-François Caulier. Ana Mauleon and Vincent Vannetelbosch: Contractually Stable Networks
66	48 2009	Massimiliano Mazzanti Susanna Mancinelli Giovanni Ponti and Nora Piva: Education Reputation or
uc .	40.2009	Network? Evidence from Italy on Migrant Workers Employability
SD	49.2009	William Brock and Anastasios Xepapadeas: General Pattern Formation in Recursive Dynamical Systems
		Models in Economics
SD	50.2009	Giovanni Marin and Massimiliano Mazzanti: Emissions Trends and Labour Productivity Dynamics Sector
		Analyses of De-coupling/Recoupling on a 1990-2005 Namea
SD	51.2009	Yoshio Kamijo and Ryo Kawasaki (lxxxv): Dynamics, Stability, and Foresight in the Shapley-Scarf Housing
		<u>Market</u>
IM	52.2009	Laura Poddi and Sergio Vergalli: <u>Does Corporate Social Responsibility Affect the Performance of Firms?</u>
SD	53.2009	Valentina Bosetti, Carlo Carraro and Massimo Tavoni: <u>Climate Change Mitigation Strategies in Fast-</u>
66	54.0000	Growing Countries: The Benefits of Early Action
GC	54.2009	Alireza Naghavi and Gianmarco I.P. Ottaviano: <u>Firm Heterogeneity, Contract Enforcement, and the Industry</u>
184	55 2000	Dynamics of Offshoring Giacome Calzelari, and Carle Scorpe: On Regulation and Competition: Proc. and Cons. of a Diversified
IIVI	33.2009	Monopolist
SD	56 2009	Valentina Bosetti Ruben Lubowski and Alexander Golub and Anil Markandva: Linking Reduced
50	00.2009	Deforestation and a Global Carbon Market: Impacts on Costs. Financial Flows, and Technological
		Innovation
IM	57.2009	Emmanuel Farhi and Jean Tirole: <u>Collective Moral Hazard, Maturity Mismatch and Systemic Bailouts</u>
SD	58.2009	Kelly C. de Bruin and Rob B. Dellink: How Harmful are Adaptation Restrictions
SD	59.2009	Rob Dellink, Michel den Elzen, Harry Aiking, Emmy Bergsma, Frans Berkhout, Thijs Dekker, Joyeeta Gupta:
		Sharing the Burden of Adaptation Financing: An Assessment of the Contributions of Countries
SD	60.2009	Stefania Tonin, Anna Alberini and Margherita Turvani: <u>The Value of Reducing Cancer Risks at Contaminated</u>
		Sites: Are More Heavily Exposed People Willing to Pay More?
SD	61.2009	Clara Costa Duarte, Maria A. Cunha-e-Sá and Renato Rosa: <u>The Role of Forests as Carbon Sinks: Land-Use</u>
66	<b>CO 0000</b>	and Carbon Accounting
	62.2009	Carlo Altomonte and Gabor Bekes: <u>Trade Complexity and Productivity</u>
GC	63.2009	Elena Bellini, Gianmarco I.P. Ottaviano, Dino Pinelli and Giovanni Prarolo: <u>Cultural Diversity and Economic</u>
SD	64 2000	<u>Performance: Evidence from European Regions</u>
30	04.2009	The Incentives to Participate in and the Stability of International Climate Coalitions: A Came-theoretic
		Analysis Using the Witch Model
IM	65.2009	John Temple Lang: Article 82 EC – The Problems and The Solution
SD	66.2009	P. Dumas and S. Hallegatte: Think Again: Higher Elasticity of Substitution Increases Economic Resilience
SD	67.2009	Ruslana Rachel Palatnik and Roberto Roson: Climate Change Assessment and Agriculture in General
		Equilibrium Models: Alternative Modeling Strategies
SD	68.2009	Paulo A.L.D. Nunes, Helen Ding and Anil Markandya: <u>The Economic Valuation of Marine Ecosystems</u>
IM	69.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders
SD	70.2009	Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the
		Emergence of Economic Agglomerations
SD	71.2009	A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: <u>Coastal Zone Management in the Mediterranean:</u>
		Legal and Economic Perspectives
GC	72.2009	Gianmarco I.P. Ottaviano and Giovanni Prarolo: <u>Cultural Identity and Knowledge Creation in Cosmopolitan</u>
		<u>Cities</u>
SD	73.2009	Erik Ansink: Self-enforcing Agreements on Water allocation
GC	/4.2009	Mario A. Maggioni, Francesca Gambarotto and T. Erika Uberti: <u>Mapping the Evolution of "Clusters": A</u>
(D	75 2000	<u>Meta-analysis</u>
SD	75.2009	INEKTARIOS ASIANIDIS: Environmental Kuznets Curves for Carbon Emissions: A Critical Survey
5D	70.2009	Joan Canton: Environmentalists' Benaviour and Environmental Policies
SD	77.2009	Christoph IVI. Kneinberger: Paying for Safety: Preferences for Mortality Risk Reductions on Alpine Roads
IM	78.2009	Chiara D'Alpaos, Michele Moretto, Paola Valbonesi and Sergio Vergalli: <u>"It is Never too late": Optimal</u>
SD	79 2000	renary for investment Delay in Fubic Frocurement Contracts Henry Tulkens and Vincent van Steenbergher "Mitigation Adaptation Suffering": In Search of the Dight Mix
50	19.2009	in the Face of Climate Change

SD	80.2009	Giovanni Bella: <u>A Search Model for Joint Implementation</u>
SD	81.2009	ZhongXiang Zhang: Multilateral Trade Measures in a Post-2012 Climate Change Regime?: What Can Be
		Taken from the Montreal Protocol and the WTO?
SD	82.2009	Antoine Dechezleprêtre, Matthieu Glachant, Ivan Hascic, Nick Johnstone and Yann Ménière: <u>Invention and</u>
<b>C</b> D		Transfer of Climate Change Mitigation Technologies on a Global Scale: A Study Drawing on Patent Data
SD	83.2009	Laszlo A. Koczy: <u>Stationary Consistent Equilibrium Coalition Structures Constitute the Recursive Core</u>
SD	84.2009	Luca Di Corato and Michele Moretto: Investing in Biogas: Timing, Technological Choice and the Value of
SD	85 2009	<u>Frexibility from inputs Mix</u> Valentina Bosetti, Enrica De Cian, Alessandra Scobbi, and Massimo Tavoni: The 2008 WITCH Model: New
30	03.2009	Model Features and Baseline
IM	86.2009	Rocco Macchiavello: Vertical Integration and Investor Protection in Developing Countries
SD	87.2009	Massimiliano Mazzanti and Antonio Musolesi: Carbon Kuznets Curves: Long-run Structural Dynamics and
		Policy Events
GC	88.2009	Gianmarco I.P. Ottaviano and Christian Volpe Martincus: <u>SMEs in Argentina: Who are the Exporters</u>
GC	89.2009	Gianpaolo Rossini and Cecilia Vergari: Input Production Joint Venture
SD	90.2009	Angelo Antoci, Simone Borghesi and Marcello Galeotti: Environmental Options and Technological
		Innovation: An Evolutionary Game Model
GC	91.2009	Cristina Cattaneo: The Decision to Migrate and Social Capital: Evidence from Albania
SD	92.2009	Valentina Bosetti and Jeffrey Frankel: Global Climate Policy Architecture and Political Feasibility: Specific
	~~ ~~~~	Formulas and Emission Targets to Attain 460 ppm CO2 Concentrations
IM	93.2009	Benno Bühler: Do International Roaming Alliances Harm Consumers?
GC	94.2009	Elena Fumagalli and Laura Fumagalli: Like Oil and Water or Chocolate and Peanut Butter? Ethnic Diversity
cc	05 2000	and Social Participation of Young People in England
UC	93.2009	Constraints: Evidence from a Pilot Study
GC	96.2009	Benedicte Apouev and Andrew E. Clark: Winning Big but Feeling no Better? The Effect of Lottery Prizes on
		Physical and Mental Health
GC	97.2009	Giovanni Gallipoli and Laura Turner: Household Responses to Individual Shocks: Disability and Labor Supply
IM	98.2009	Felix Bierbrauer: On the Legitimacy of Coercion for the Financing of Public Goods
IM	99.2009	Ángel L. López and Patrick Rey: Foreclosing Competition through Access Charges and Price Discrimination
SD	100.2009	Massimo Tavoni and Bob van der Zwaan: Nuclear versus Coal plus CCS: A Comparison of Two Competitive
		Base-load Climate Control Options
SD	101.2009	Dinko Dimitrov and Clemens Puppe: Non-bossy Social Classification
GC	102.2009	Ngo Van Long, Antoine Soubeyran and Raphael Soubeyran: <u>The Pace of Technology Transfer in Anticipation</u>
	102 0000	of Joint Venture Breakup
IM	103.2009	Mariassunta Giannetti and Andrei Simonov: <u>On the Real Effects of Bank Bailouts: Micro-Evidence from</u>
SD	10/ 2000	Japan Xavier Pautrel: Time-separable Utility Leisure and Human Capital Accumulation: What New Implications for
30	104.2009	the Environment-Growth Nexus?
SD	105.2009	Johannes Bollen, Sebastiaan Hers and Bob van der Zwaan: An Integrated Assessment of Climate Change, Air
		Pollution, and Energy Security Policy
SD	106.2009	Nicholas Stern: Imperfections in the Economics of Public Policy, Imperfections in Markets, and Climate
		<u>Change</u>
GC	107.2009	Giorgio Bellettini, Carlotta Berti Ceroni and Giovanni Prarolo: Political Persistence, Connections and
	100 0000	Economic Growth
IM	108.2009	Xavier Boutin, Giacinta Cestone, Chiara Fumagalli, Giovanni Pica and Nicolas Serrano-Velarde: <u>The Deep-</u>
SD	109 2009	<u>Pocket Effect of Internal Capital Markets</u> Michael Finus, Bianca Pundshagen and Johan Evolutions: Simulating a Sequential Coalition Formation
50	109.2009	Process for the Climate Change Problem: First Come but Second Served?
SD	110.2009	Barry Anderson, Corrado DiMaria and Frank Convery: Abatement and Allocation in the Pilot Phase of the
		<u>EU ETS</u>
SD	111.2009	Xavier Pautrel: Health-enhancing activities and the environment: How competition for resources make the
		environmental policy beneficial
IM	112.2009	Yishay Yafeh and Beni Lauterbach: Long Term Changes in Voting Power and Control Structure Following the
<b>CD</b>	112 0000	Unification of Dual Class Shares
50	113.2009	Julien Chevallier and Benoît Sévi: <u>On the realized volatility of the ECX CO2 emissions 2008 futures contract:</u>
SD	114 2000	aistribution, aynamics and forecasting Frik Ansink and Hans-Datar Waikard: Secuential charing rules for river charing problems
GC	115.2009	Gregory Corcos, Massimo Del Gatto, Giordano Mion, and Gianmarco LP. Ottaviano: Productivity and Firm
		Selection: Quantifying the "New" Gains from Trade

(lxxxv) This paper has been presented at the 14th Coalition Theory Network Workshop held in Maastricht, The Netherlands, on 23-24 January 2009 and organised by the Maastricht University CTN group (Department of Economics, http://www.feem-web.it/ctn/12d\_maa.php).